Author's response to reviews

Title: Habitual Prospective Memory in Schizophrenia

Authors:

Dr Brita Elvevag (elvevaab@intra.nimh.nih.gov)
Elizabeth A Maylor (E.A.Maylor@warwick.ac.uk)
Abigail L Gilbert (gilberta@mail.nih.gov)

Version: 4 Date: 21 Jul 2003

To the Editor,

Thank you for your recent email on our paper titled "Habitual prospective memory in schizophrenia".

As suggested, we now emphasize in the "Results" section the problem of controls making few omission errors and alert the reader to the fact that this will be discussed further in the "Discussion" section. We have done this by inserting the following sentence in the second paragraph of the "Results" section: "We note that this low rate of omission errors in controls reduces the opportunity to report a missed action as having been performed (see later Discussion)."

Once again, we are grateful for the helpful reviews that we received on our manuscript, and we very much hope that this revision is suitable for publication in BMC Psychiatry.

We look forward to hearing from you.

Yours sincerely,

Brita Elvevag, Ph.D.